Author's response to reviews

Title: Case report of an incidental giant oncocytoma.

Authors:

Anastasios AA Anastasiadis (dranastasiadis@yahoo.gr)
Georgios GD Dimitriadis (gdimit@auth.gr)
Dimitrios DR Radopoulos (dradopoulos@hotmail.gr)

Version: 2 Date: 10 February 2010

Author's response to reviews:

The abstract was restructured. The title was changed. The main part was restructured (all according to the instructions provided).